**Author’s response to reviews**

**Title:** Integration opportunities for HIV and family planning services in Addis Ababa, Ethiopia: an organizational network analysis

**Authors:**

- James C Thomas (jim.thomas@unc.edu)
- Heidi Reynolds (heidi_reynolds@unc.edu)
- Christine Bevc (bevc@email.unc.edu)
- Ademe Tsegaye (atsegaye@msh.org)

**Version:** 2  **Date:** 19 July 2013

**Author’s response to reviews:**

As requested by the journal, a statement describing informed consent was added to the Methods section.